\documentclass[twoside]{article}
%\usepackage{graphics}
%\usepackage{graphicx}
\usepackage{latexsym}
\usepackage{amssymb}
\usepackage{amsmath}
\usepackage{BSPMsample}
\begin{document}
\begin{thebibliography}{99}

\bibitem{ArasandSonmezandCakalliAnapproachtosoftfunctions} C.G. Aras, A. Sonmez, H. Cakalli, {\em An approach to soft functions}, J. Math. Anal. \textbf{8}, 2, 129-138, (2017).
\bibitem{BorOnGeneralizedAbsoluteCesaroSummability} H. Bor, {\em On Generalized Absolute Cesaro Summability}, An. Stiint. Univ. Al. I. Cuza Iasi. Mat. (N.S.)  \textbf{57}, 2, 323-328, (2011). DOI: 10.2478/v10157-011-0029-9
\bibitem{BrahaandCakalliAnewtypecontinuityforrealfunctions} Naim L. Braha, H. Cakalli, {\em A new type continuity for real functions}, J. Math. Anal. \textbf{7},  6, 68-76, (2016).
%J. Math. Anal. Journal of Mathematical Analysis. Ilirias Publ., Pri^stina. ISSN 2217-3412.
\bibitem{BurtonandColemanQuasiCauchysequences} D. Burton, and J. Coleman, {\em Quasi-Cauchy Sequences}, Amer. Math. Monthly \textbf{117}, 4, 328-333, (2010).
\bibitem{CakalliAvariationonarithmeticcontinuity} H. Cakalli, {\em A variation on arithmetic continuity}, Bol. Soc. Paran. Mat. \textbf{35}, 3, 195-202, (2017).
\bibitem{CakalliAvariationonstatisticalwardcontinuity} H. Cakalli, {\em A variation on statistical ward continuity}, Bull. Malays. Math. Sci. Soc.  \textbf{\textbf{40}} (2017) 1701-1710. https://doi.org/10.1007/s40840-015-0195-0
%%%H. Cakalli, {\em A Variation on Statistical Ward Continuity}, Bull. Malays. Math. Sci. Soc. (2015). https://doi.org/10.1007/s40840-015-0195-0
%%%H. Cakalli, A variation on statistical ward continuity, Bull. Malays. Math. Sci. Soc.  \textbf{\textbf{40}} (2017) 1701-1710. https://doi.org/10.1007/s40840-015-0195-0
%Cakalli, H. Bull. Malays. Math. Sci. Soc. (2015). https://doi.org/10.1007/s40840-015-0195-0
\bibitem{CakalliandAlbayrakNewtypecontinuitiesviaAbelconvergence} H. Cakalli and M. Albayrak, {\em New Type Continuities via Abel Convergence}, The Scientific World Journal Volume 2014, Article ID 398379, 6 pages, (2014). http://dx.doi.org/10.1155/2014/398379
\bibitem{CakalliandArasandSonmezLacunarystatisticalwardcontinuity} H. \c{C}akalli, C.G. Aras, and A. Sonmez,  {\em Lacunary statistical ward continuity}, AIP Conf. Proc. \textbf{1676}, Article Number: 020042, (2015). doi: 10.1063/1.4930468
\bibitem{CakalliandHazarikaIdealquasiCauchysequences} H. Cakalli and B. Hazarika, {\em Ideal quasi-Cauchy sequences}, J. Inequal. Appl. 2012, 2012:234, (2012).  doi:10.1186/1029-242X-2012-234
\bibitem{CakalliandKaplanAvariationonstronglylacunarywardcontinuityJournalofMathematicalAnalysis}	H. Cakalli and H. Kaplan, {\em A variation on strongly lacunary ward continuity}, J. Math. Anal. \textbf{7}, 3, 13-20, (2016).
\bibitem{CakalliandMucukLacunarystatisticallyupwardanddownwardhalfquasiCauchysequencesJofAnalysis} H. Cakalli and O. Mucuk, {\em Lacunary statistically upward and downward half quasi-Cauchy sequences}, J. Math. Anal. \textbf{7}, 2, 12-23, (2016).
\bibitem{CasertaandKocinacOnstatisticalexhaustiveness} A. Caserta, and Ljubisa. D. R. Ko\v cinac,  {\em On statistical exhaustiveness}, Appl. Math. Lett. \textbf{25}, 10, 1447-1451, (2012).
\bibitem{ConnorandGrosseSequentialdefinitionsofcontinuityforrealfunctions} J.Connor, K.-G.Grosse-Erdmann, {\em Sequential definitions of continuity for real functions}, Rocky Mountain J. Math. \textbf{33}, 1, 93-121, (2003).
\bibitem{CakalliSlowlyoscillatingcontinuity} H. \c{C}akall\i, {\em Slowly oscillating continuity}, Abstr. Appl. Anal. Hindawi Publ. Corp. New York,
ISSN 1085-3375, Volume 2008, Article ID 485706, (2008). doi:10.1155/2008/485706
\bibitem{CakalliSequentialdefinitionsofcompactness} H. \c{C}akall\i, {\em Sequential definitions of compactness}, Appl. Math. Lett. \textbf{21}, 6, 594-598, (2008).
\bibitem{CakalliAstudyonstatisticalconvergence}  H. \c{C}akall\i,  {\em A study on statistical convergence}, Funct. Anal. Approx. Comput.   \textbf{1}, 2,  19-24, (2009).
\bibitem{CakalliForwardcontinuity}  H. \c{C}akall\i,  {\em Forward continuity},  J. Comput. Anal. Appl. \textbf{13}, 2, 225-230, (2011).
\bibitem{CakalliOnDeltaquasislowlyoscillatingsequences} H. \c{C}akall\i, {\em On $\Delta$-quasi-slowly oscillating sequences}, Comput. Math. Appl. \textbf{62}, 9, 3567-3574, (2011).
\bibitem{CakalliStatisticalquasiCauchysequences}  H. \c{C}akall\i,  {\em Statistical quasi-Cauchy sequences}, Math. Comput. Modelling, \textbf{54}, no. 5-6, 1620-1624, (2011).
\bibitem{CakalliDeltaquasiCauchysequences}  H. \c{C}akall\i,  {\em $\delta$ quasi-Cauchy sequences}, Math. Comput. Modelling, \textbf{53}, no. 1-2, 397-401, (2011).
\bibitem{CakalliStatisticalwardcontinuity} H. \c{C}akall\i, {\em Statistical ward continuity}, Appl. Math. Lett. \textbf{24}, 10, 1724-1728, (2011).
\bibitem{CakalliOnGcontinuity} H. \c{C}akall\i,  {\em On $G$-continuity}, Comput. Math. Appl. \textbf{61}, 2, 313-318, (2011).
\bibitem{CakalliSequentialdefinitionsofconnectedness} H. \c{C}akall\i,  {\em Sequential definitions of connectedness}, Appl. Math. Lett., \textbf{25}, 3, 461-465, (2012).
\bibitem{CakalliNthetawardcontinuity} H. \c{C}akall\i,  {\em N-theta-Ward continuity}, Abstr. Appl. Anal. Hindawi Publ. Corp., New York, Volume \textbf{2012}, Article ID 680456, 8 pp, (2012). doi:10.1155/2012/680456.
\bibitem{CakalliAvariationonwardcontinuity} H. \c{C}akall\i,  {\em A variation on ward continuity}, Filomat, \textbf{27}, 8, 1545-1549, (2013). DOI: 10.2298/FIL1308545C
\bibitem{CakalliVariationsonquasiCauchysequences} H. \c{C}akall\i,  {\em Variations on quasi-Cauchy sequences}, Filomat, \textbf{29}, 1, 13-19, (2015).
 %H. C� akall�, Variations on quasi-Cauchy sequences, Filomat, 29 (1) (2015), 13-19.
\bibitem{CakalliUpwardanddownwardstatisticalcontinuities} H. \c{C}akall\i,  {\em Upward and downward statistical continuities}, Filomat, \textbf{29}, 10, 2265-2273, (2015).
\bibitem{CakalliandSonmezandGencOnanequivalenceoftopologicalvectorspacevaluedconemetricspacesandmetricspaces}  H. \c{C}akall\i, A. Sonmez, and \c{C}. Gen\c{c}, {\em On an equivalence of topological vector space valued cone metric spaces and metric spaces}, Appl. Math. Lett. \textbf{25}, 3, 429-433,  (2012).
%; http://dx.doi.org/10.1063/1.4930468
%%%\bibitem{CakalliandArasandSonmezLacunarystatisticalwardcontinuity} H. \c{C}akalli, C.G. Aras, A. Sonmez, On lacunary statistically quasi-Cauchy sequences, submitted and Algerian-Turkish International Days on Mathematics�2013, ATIM 2013, September 12-19, \textbf{2013} arXiv:1102.1531
\bibitem{CakalliGCanakPnsabsolutealmostconvergentsequences} H. Cakalli, and G. Canak, {\em $(P_{n} ,s)$-absolute almost convergent sequences}, Indian J. Pure Appl. Math. \textbf{28}, 4, 525-532, (1997).
\bibitem{CakalliandPratulFuzzycompactnessviasummability} H. \c{C}akalli and Pratulananda Das, {\em Fuzzy compactness via summability}, Appl. Math. Lett. \textbf{22}, 11, 1665-1669, (2009).
\bibitem{CakalliandKaplanAstudyonNthetaquasiCauchysequences} H. \c{C}akall\i, and H. Kaplan, {\em A study on N-theta quasi-Cauchy sequences},  Abstr. Appl. Anal., Hindawi Publ. Corp., New York, Volume \textbf{2013}, Article ID 836970 Article ID 836970, 4 pages,(2013). doi:10.1155/2013/836970
\bibitem{CakalliandKaplanAvariationonlacunarystatisticalquasiCauchysequencesCommunications} H. Cakalli, and H. Kaplan, {\em A variation on lacunary statistical quasi Cauchy sequences}, Communications Faculty of Sciences University of Ankara-Series A1 Mathematics and Statistics, \textbf{66}, 2, 71-79, (2017). 10.1501/Commua1 0000000802
    %DOI: 10.1501/Commua1_0000000802 Published: 2017
%A VARIATION ON LACUNARY STATISTICAL QUASI CAUCHY SEQUENCES COMMUNICATIONS FACULTY OF SCIENCES UNIVERSITY OF ANKARA-SERIES A1 MATHEMATICS AND STATISTICS Volume: 66  Issue: 2  Pages: 71-79 DOI: 10.1501/Commua1_0000000802 Published: 2017
\bibitem{CakalliandKhanSummabilityintopologicalspaces} H. \c{C}akalli, and M.K. Khan, {\em Summability in topological spaces}, Appl. Math. Lett. \textbf{24}, 348-352,(2011).
\bibitem{CakalliandMucukconnectednessviaasequentialmethod} H. Cakalli and O. Mucuk, {\em On connectedness via a sequential method}, Revista de la Union Matematica Argentina, \textbf{54}, 2, 101-109, (2013).
%Citation: AIP Conference Proceedings 1759, 020038 (2016); doi: 10.1063/1.4959652
\bibitem{CakalliandPattersonFunctionspreservingslowlyoscillatingdoublesequences} H. \c{C}akall\i\;, R.F. Patterson, {\em Functions preserving slowly oscillating double sequences}, An. Stiint. Univ. Al. I. Cuza Iasi. Mat. (N.S.) \textbf{62}, 2, vol. 2. 531-536, (2016). http://www.math.uaic.ro/~annalsmath/pdf-uri%20anale/F2-2(2016)/Cakalli-Patterson.pdf
%Cakalli, Huseyin; Patterson, Richard F. Functions preserving slowly oscillating double sequences. An. �tiint. Univ. Al. I. Cuza Ia�i. Mat. (N.S.) 62 (2016), no. 2, vol. 2, 531--536. MR3680231
%An. S�tiint�. Univ. Al. I. Cuza Ia�si Mat. (N.S.) Tomul LXII, 2016, f. 2, vol. 2
\bibitem{CakalliandSonmezSlowlyoscillatingcontinuityinabstractmetricspaces} H. Cakalli, A. Sonmez,  {\em Slowly oscillating continuity in abstract metric spaces} Filomat, \textbf{27}, 5, 925-930, (2013).
\bibitem{CakalliandSonmezandAraslamdastatisticallywardcontinuity}  H. \c{C}akall\i, A. S\"onmez , and  \c{C}.G. Aras, {\em $\lambda$-statistically ward continuity}, An. Stiint. Univ. Al. I. Cuza Iasi. Mat. Tomul LXII, 2017, Tom LXIII, f. 2, 308-3012, (2017). DOI: 10.1515/aicu-2015-0016
\bibitem{CakalliandTaylanOnabsolutelyalmostconvergence} H. \c{C}akall\i, and E.Iffet Taylan, {\em On Absolutely Almost Convergence}, An. Stiint. Univ. Al. I. Cuza Iasi. Mat. (N.S.), Tom LXIII, f. 1  1-6, (2017). Doi: 10.2478/aicu-2014-0032
\bibitem{CanakandDikNewtypesofcontinuities} I. Canak and M. Dik, {\em New types of continuities}, Abstr. Appl. Anal. 2010, Article ID 258980, 6 pages, (2010).
\bibitem{DjurcicandKocinacandZizovicDoublesequencesandselections} D. Djurcic, Ljubisa D.R. Kocinac, M.R. Zizovic, {\em Double sequences and selections}, Abstr. Appl. Anal.  Art. ID 497594, 6 pp, (2012).
\bibitem{ErdemandArasandCakalliandSonmezSoftmatricesonsoftmultisetsinanoptimaldecisionprocess} A.E. Coskun, C.G Aras, H. Cakalli, and A. Sonmez, {\em Soft matrices on soft multisets in an optimal decision process}, AIP Conference Proceedings, \textbf{1759}, 1,  020099 (2016); doi: 10.1063/1.4959713
\bibitem{FDikMDikandCanak} F. Dik, M. Dik, and I. Canak, {\em Applications of subsequential Tauberian theory to classical Tauberian theory}, Appl. Math. Lett., \textbf{20}, 8,  946-950, (2007).
\bibitem{CanakandDik} I. Canak, and M. Dik, {\em New Types of Continuities}, Abstr. Appl. Anal., Hindawi Publ. Corp., New York, ISSN 1085-3375, Volume 2010, Article ID 258980, (2010).  doi:10.1155/2010/258980
\bibitem{Fridy} J.A. Fridy, {\em On statistical convergence}, Analysis, \textbf{5}, 4, 301-313, (1985).
\bibitem{KaplanandCakalliVariationsonstronglylacunaryquasiCauchysequencesAIP} H. Kaplan, H. Cakalli, {\em Variations on strongly lacunary quasi Cauchy sequences}, AIP Conf. Proc. \textbf{1759}, Article Number: 020051, (2016). doi: http://dx.doi.org/10.1063/1.4959665
\bibitem{KaplanandCakalliVariationsonstronglacunaryquasiCauchysequencesJNonlinearSciAppl} H. Kaplan, H. Cakalli, {\em Variations on strong lacunary quasi-Cauchy sequences}, J. Nonlinear Sci. Appl. \textbf{9}, 4371-4380, (2016).
%\bibitem{KaplanandCakalliVariationsonstronglylacunaryquasiCauchysequencesAIP} H. Kaplan, H. Cakalli, Variations on strongly lacunary quasi Cauchy sequences, AIP Conf. Proc. \textbf{1759} (2016) Article Number: 020051
\bibitem{KocinacSelectionpropertiesinfuzzymetricspaces}  Ljubisa D.R. Kocinac, {\em Selection properties in fuzzy metric spaces}, Filomat, \textbf{26}, 2, 305-312, (2012).
\bibitem{MucukandCakalliGsequentiallyconnectednessfortopologicalgroupswithoperations} O. Mucuk and H. Cakalli, {\em G-sequentially connectedness  for topological groups with operations}, AIP Conference Proceedings, \textbf{1759}, 020038, (2016). doi: 10.1063/1.4959652
\bibitem{MucukandSahanOnGsequentialcontinuity} O. Mucuk, T. \c{S}ahan� {\em On $G$-Sequential Continuity}, Filomat, \textbf{28}, 6, 1181-1189, (2014).
DOI 10.2298/FIL1406181M
\bibitem{NatarajanClassicalsummabilitytheory} P.N. Natarajan, {\em Classical Summability Theory}, Springer Nature Singapore Pte Ltd. 130 pages, (2017).  doi: 10.1007/978-981-10-4205-8
\bibitem{OzarslanandYildizAnewstudyontheabsolutesummabilityfactorsofFourierseries} H.S. Ozarslan, and \c{S}. Y\i ld\i z, {\em A new study on the absolute summability factors of Fourier series}, J. Math. Anal. 7, 1, 31-36, (2016).
\bibitem{PalandSavasandCakalliIconvergenceonconemetricspaces}	S.K. Pal,  E. Savas, and H. Cakalli, {\em $I$-convergence on cone metric spaces}, Sarajevo J. Math. \textbf{9}, 85-93, (2013).
\bibitem{PattersonandCakalliQuasiCauchydoublesequences} R.F. Patterson and H. Cakalli, {\em Quasi Cauchy double sequences}, Tbilisi Math. J., \textbf{8}, 2, 211-219,  (2015).
\bibitem{PattersonandSavasAsymptoticequivalenceofdoublesequences} Richard F. Patterson, and E. Savas, {\em Asymptotic equivalence of double sequences}, Hacet. J. Math. Stat.  \textbf{41}, 4, 487-497, (2012).
\bibitem{SavasandCakalliIdealstatisticalquasiCauchysequences} E. Savas, H. \c{C}akall\i, {\em Ideal statistically quasi Cauchy sequences}, AIP Conf. Proc. \textbf{1759}, 020057, (2016). doi: 10.1063/1.4959671
    %International Conference on Recent Advances in Pure and Applied Mathematics,  (ICRAPAM 2014) 6-9 November 2014, (2014). Antalya, Turkey.
% AIP Conf. Proc. \textbf{1759}, 020057, (2016). doi: 10.1063/1.4959671
\bibitem{SonmezOnparacompactnessinconemetricspaces} A. Sonmez, {\em On paracompactness in cone metric spaces}, Appl. Math. Lett. \textbf{23}, 494-497, (2010).
\bibitem{Vallin} R.W. Vallin, {\em Creating slowly oscillating sequences and slowly oscillating continuous functions,  With an appendix by Vallin and H. Cakalli}, Acta Math. Univ. Comenianae, \textbf{25}, 1, 71-78, (2011).
\bibitem{YayingandHazarikaOnarithmeticcontinuity} T. Yaying and B. Hazarika, {\em On arithmetic continuity}, Bol. Soc. Paran. Mat. \textbf{35}, 1, (2017), 139-145, (2017).
\bibitem{YayingandHazarikandCakalliNewresultsinquasiconemetricspaces} T. Yaying, B. Hazarika, H. Cakalli, {\em New results in quasi cone metric spaces}, J. Math. Computer Sci. \textbf{16}, 435-444, (2016).
\bibitem{YildizAnewtheoremonlocalpropertiesoffactoredFourierseries} S. Yildiz, {\em A new theorem on local properties of factored Fourier series}, Bull. Math. Anal. Appl. \textbf{8}, 2, 1-8, (2016).
\bibitem{YildizOnAbsoluteMatrixSummabilityFactorsofInfiniteSeriesandFourierSeries}   \c{S}. Y\i ld\i z, {\em On Absolute Matrix Summability Factors of Infinite Series and Fourier Series}, Gazi University Journal of Science, \textbf{30}, 1, 363-370, (2017).
\bibitem{YildizIstatistikselboslukludelta2quasiCauchydizileri} \c{S}. Y\i ld\i z, {\em \.{I}statistiksel bo\c{s}luklu delta 2 quasi Cauchy dizileri}, Sakarya University Journal of Science, \textbf{21}, 6, (2017).
%Sakarya �niversitesi Fen Bilimleri Enstitusu Dergisi,
%\bibitem{LionSIAM}  Lions, J. L.,
%{\em Exact Controllability, Stabilizability, and Perturbations for
%Distributed Systems}, Siam Rev. 30, 1-68, (1988).
%\bibitem{Da71} C. M. Dafermos, C. M., {\it An abstract Volterra equation with application
%to linear viscoelasticity}. J. Differential Equations 7, 554-589,
%(1970).
\end{thebibliography}


\title[Variations on statistical quasi Cauchy sequences ]
{Variations on statistical quasi Cauchy sequences }
%between [] goes the title that will appear on the top of every
%odd page, between {} goes the real title.
%\thanks{...} put your grant

\author[H. Cakalli]{Huseyin Cakalli}
%between [] goes the authors that will appear on the top of every
%even page, between {} goes the real author.

\address{Huseyin Cakalli\\Maltepe University, Department of Mathematics, \\
Graduate School of Science and Engineering, Marmara E\u{g}\.{I}t\.{I}m K\"oy\"u, TR 34857, Maltepe, \.{I}stanbul \\ Turkey\\
\email: hcakalli@altepe.edu.tr; huseyincakalli@altepe.edu.tr; hcakalli@gmail.com}

\maketitle

\begin{abstract}
In this paper, we introduce a concept of statistically $p$-quasi-Cauchyness of a real sequence in the sense that a sequence $(\alpha_{k})$ is statistically $p$-quasi-Cauchy if $\lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq n: |\alpha_{k+p}-\alpha_{k}|\geq{\varepsilon}\}|=0$ for each $\varepsilon>0$. A function $f$ is called statistically $p$-ward continuous on a subset $A$ of the set of real umbers $\mathbb{R}$ if it preserves statistically $p$-quasi-Cauchy sequences, i.e. the sequence $f(\textbf{x})=(f(\alpha_{n}))$ is statistically $p$-quasi-Cauchy whenever $\boldsymbol\alpha=(\alpha_{n})$ is a statistically $p$-quasi-Cauchy sequence of points in $A$. It turns out that a real valued function $f$ is uniformly continuous on a bounded subset $A$ of $\mathbb{R}$ if there exists a positive integer $p$ such that $f$ preserves statistically $p$-quasi-Cauchy sequences of points in $A$.
\end{abstract}
%%put here Key words

\keywords sequences, series, real functions, continuity, compactness.


\tableofcontents



%% put here the subject class
\2000mathclass{Primary: 40A05; Secondaries: 40A30, 26A15}


\section{Introduction}\label{sec1}
Throughout this paper, $\mathbb{N}$, and $\mathbb{R}$ will denote the set of positive integers, and the set of real numbers, respectively. $p$ will always be a fixed element of $\mathbb{N}$. The boldface letters such as $\boldsymbol\alpha$, $\boldsymbol\beta$, $\boldsymbol\zeta$ will be used for sequences $\boldsymbol\alpha=(\alpha_{n})$, $\boldsymbol\beta=(\beta_{n})$, $\boldsymbol\zeta=(\zeta_{n})$, ... of points in $\mathbb{R}$. A function $f:\mathbb{R} \longrightarrow \mathbb{R}$ is continuous if and only if it preserves convergent sequences. Using the idea of continuity of a real function in this manner, many kinds of continuities were introduced and investigated, not all but some of them we recall in the following: ward continuity (\cite{CakalliForwardcontinuity}, \cite{BurtonandColemanQuasiCauchysequences}), $p$-ward continuity (\cite{CakalliVariationsonquasiCauchysequences}),  $\delta$-ward continuity (\cite{CakalliDeltaquasiCauchysequences}), $\delta^{2}$-ward continuity (\cite{BrahaandCakalliAnewtypecontinuityforrealfunctions}), statistical ward continuity,  (\cite{CakalliStatisticalwardcontinuity}), $\lambda$-statistical ward continuity (\cite{CakalliandSonmezandAraslamdastatisticallywardcontinuity}), $\rho$-statistical ward continuity (\cite{CakalliAvariationonstatisticalwardcontinuity}), slowly oscillating continuity  (\cite{CakalliSlowlyoscillatingcontinuity,Vallin, CakalliandSonmezSlowlyoscillatingcontinuityinabstractmetricspaces}), quasi-slowly oscillating continuity (\cite{CanakandDik}), $\Delta$-quasi-slowly oscillating continuity (\cite{CakalliOnDeltaquasislowlyoscillatingsequences}), arithmetic continuity (\cite{YayingandHazarikaOnarithmeticcontinuity}, \cite{CakalliAvariationonarithmeticcontinuity}), upward and downward statistical continuities (\cite{CakalliUpwardanddownwardstatisticalcontinuities}), lacunary statistical ward continuity (\cite{CakalliandArasandSonmezLacunarystatisticalwardcontinuity}), $\delta$ lacunary statistical ward continuity (\cite{CakalliandKaplanAvariationonlacunarystatisticalquasiCauchysequencesCommunications}), $\delta^{2}$ lacunary statistical ward continuity (\cite{YildizIstatistikselboslukludelta2quasiCauchydizileri}), ideal ward continuity (\cite{CakalliandHazarikaIdealquasiCauchysequences, CakalliAvariationonwardcontinuity}), ideal statistical ward continuity (\cite{SavasandCakalliIdealstatisticalquasiCauchysequences}), $N_{\theta}$-ward continuity (\cite{CakalliNthetawardcontinuity}, \cite{CakalliandKaplanAstudyonNthetaquasiCauchysequences}, \cite{CakalliandKaplanAvariationonstronglylacunarywardcontinuityJournalofMathematicalAnalysis}, \cite{KaplanandCakalliVariationsonstronglacunaryquasiCauchysequencesJNonlinearSciAppl}, \cite{KaplanandCakalliVariationsonstronglylacunaryquasiCauchysequencesAIP}), $N_{\theta}$-$\delta$-ward continuity (\cite{CakalliandKaplanAvariationonstronglylacunarywardcontinuityJournalofMathematicalAnalysis}), and Abel continuity (\cite{CakalliandAlbayrakNewtypecontinuitiesviaAbelconvergence}), which enabled some authors to obtain interesting results.

The purpose of this paper is to introduce statistically $p$-quasi-Cauchy sequences, and prove interesting theorems.

\section{Variations on statistical ward compactness}
The concept of a Cauchy sequence involves far more than that the distance between successive terms is tending to $0$ and statistically tending to zero, and more generally speaking, than that the distance between $p$-successive terms is statistically tending to zero, by $p$-successive terms we mean $\alpha_{k+p}$ and $\alpha_{k}$. Nevertheless, sequences which satisfy this weaker property are interesting in their own right.

Before giving our main definition we recall basic concepts. A sequence $(\alpha_{n})$ is called quasi Cauchy if $\lim_{n\rightarrow\infty}\Delta \alpha_{n}=0$, where $\Delta \alpha_{n}=\alpha_{n+1}-\alpha_{n}$ for each $n\in{\mathbb{N}}$ (\cite{BurtonandColemanQuasiCauchysequences}, \cite{CakalliForwardcontinuity}). The set of all bounded quasi-Cauchy sequences is a closed subspace of the space of all bounded sequences with respect to the norm defined for bounded sequences (\cite{NatarajanClassicalsummabilitytheory}). A sequence $(\alpha_{k})$ of points in $\mathbb{R}$ is slowly oscillating if $ \lim_{\lambda \rightarrow 1^{+}}\overline{\lim}_{n}\max _{n+1\leq k\leq [\lambda n]} |\alpha_{k}-\alpha_{n}| =0$, where $[\lambda n]$ denotes the integer part of $\lambda n$ (\cite{FDikMDikandCanak}). A sequence $(\alpha_{k})$ is quasi-slowly oscillating if $(\Delta \alpha_{k})$ is slowly oscillating. A sequence $(\alpha_{n})$ is called statistically convergent to a real number $L$ if $\lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq n: |\alpha_{k}-L|\geq{\varepsilon}\}|=0$ for each $\varepsilon>0$ (\cite{Fridy}, \cite{CakalliAstudyonstatisticalconvergence}, \cite{CakalliandKhanSummabilityintopologicalspaces}, and \cite{CasertaandKocinacOnstatisticalexhaustiveness}). Recently in \cite{CakalliVariationsonquasiCauchysequences} it was proved that a real valued function is uniformly continuous whenever it is $p$-ward continuous on a bounded subset of $\mathbb{R}$. Now we introduce the concept of a statistically $p$-quasi-Cauchy sequence.
\begin{Defn} A sequence $(\alpha_{k})$ of points in $\mathbb{R}$ is called statistically $p$-quasi-Cauchy if $st-\lim_{k\rightarrow\infty} \Delta_{p} \alpha_{k}=0$, ie.  $\lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq n: |\Delta_{p} \alpha_{k}|\geq{\varepsilon}\}|=0$ for each $\varepsilon>0$, where $\Delta_{p} \alpha_{k}=\alpha_{k+p}-\alpha_{k}$ for every $k\in{\mathbb{N}}$.
\end{Defn}
We will denote the set of all statistically $p$-quasi-Cauchy sequences by $\Delta^{s}_{p}$ for each $p\in{\mathbb{N}}$. The sum of two statistically $p$-quasi-Cauchy sequences is statistically $p$-quasi-Cauchy, the product of a statistically $p$-quasi-Cauchy sequence and a constant real number is statistically $p$-quasi-Cauchy, so that the set of all statistically $p$-quasi-Cauchy sequences  $\Delta^{s}_{p}$ is a vector space. We note that a  sequence is statistically quasi-Cauchy when $p=1$, i.e. statistically $1$-quasi-Cauchy sequences are statistical quasi-Cauchy sequences. It follows from the inclusion \\$\{k\leq n: |\alpha_{k+p}-\alpha_{k}| \geq\varepsilon\}\subseteq\\ \subseteq  \{k\leq n: |\alpha_{k+p}-\alpha_{k+p-1}| \geq \frac{\varepsilon}{p}\} \cup \{k\leq n: |\alpha_{k+p-1}-\alpha_{k+p-2}| \geq \frac{\varepsilon}{p}\}\cup ...\\ \cup \{k\leq n: |\alpha_{k+2}-\alpha_{k+1}| \geq \frac{\varepsilon}{p}\} \cup \{k\leq n: |\alpha_{k+1}-\alpha_{k}| \geq \frac{\varepsilon}{p}\}$\\
that any statistically quasi-Cauchy sequence is also statistically $p$-quasi-Cauchy, but the converse is not always true as it can be seen by considering the the sequence $(\alpha_{k})$ defined by $(\alpha_{k})=(0,1,0,1,...,0,1,...)$ is $2$-statistically quasi Cauchy which is not statistically quasi Cauchy. More examples can be seen  in \cite[Section 1.4]{NatarajanClassicalsummabilitytheory}. It is clear that any Cauchy sequence is in $\bigcap^{p=1}_{\infty}\Delta^{s}_{p}$, so that each $\Delta^{s}_{p}$ is a sequence space containing the space $\mathcal{C}$ of Cauchy sequences. It is also to be noted that $\mathcal{C}$ is a proper subset of $\Delta^{s}_{p}$ for each $p\in{\mathbb{N}}$.

\begin{Defn} A subset $A$ of $\mathbb{R}$ is called statistically $p$-ward compact if any sequence of points in $A$ has a statistically $p$-quasi-Cauchy subsequence.
\end{Defn}
We note that this definition of statistically $p$-ward compactness cannot be obtained by any summability matrix in the sense of \cite{CakalliSequentialdefinitionsofcompactness} (see also \cite{ConnorandGrosseSequentialdefinitionsofcontinuityforrealfunctions}, and \cite{CakalliOnGcontinuity}).

Since any statistically quasi-Cauchy sequence is statistically $p$-quasi-Cauchy we see that any ward compact subset of $\mathbb{R}$ is statistically $p$-ward compact for any $p\in{\mathbb{N}}$. A finite subset of $\mathbb{R}$ is statistically $p$-ward compact, the union of finite number of statistically $p$-ward compact subsets of $\mathbb{R}$ is statistically $p$-ward compact, and the intersection of any family of statistically $p$-ward compact subsets of $\mathbb{R}$ is statistically $p$-ward compact. Furthermore any subset of a statistically $p$-ward compact set of $\mathbb{R}$ is statistically $p$-ward compact and any bounded subset of $\mathbb{R}$ is statistically $p$-ward compact. These observations above suggest to us the following.
\begin{thm} \label{ThmEisboundediffitispwardcompact}  A subset $A$ of $\mathbb{R}$ is bounded if and only if there exists a $p\in{\mathbb{N}}$ such that $A$ is statistically $p$-ward compact.
\end{thm}
\begin{pf}
The bounded subsets of $\mathbb{R}$ are statistically $p$-ward compact, since any bounded sequence of points in a bounded subset of $\mathbb{R}$ is bounded and  any bounded sequence has a convergent subsequence which is statistically $p$-quasi-Cauchy for any $p\in{\mathbb{N}}$. To prove the converse, suppose that $A$ is not bounded. If it is unbounded above, pick an element $\alpha_{1}$ of $A$ greater than $p$. Then we can find an element $\alpha_{2}$ of $A$ such that $\alpha_{2}>2p+\alpha_{1}$. Similarly, choose an element $\alpha_{3}$ of $A$ such that $\alpha_{3}>3p+\alpha_{2}$. So we can construct a sequence $(\alpha_{j})$ of numbers in  $A$ such that  $\alpha_{j+1}>(j+1)p+\alpha _{j}$ for each $j\in{\mathbb{N}}$. Let $(\alpha _{j_{k}})$ be any subsequence of $(\alpha _{j})$. Since $\{k\leq n: |\alpha _{j_{k+1}}-\alpha _{j_{k}}|\geq\frac{p}{2}\}=\{1, 2, ..., n\}$, it follows that
$\lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq n: |\alpha _{j_{k+1}}-\alpha _{j_{k}}|\geq\frac{p}{2}\}|=1\neq{0}$. This means that $(\alpha _{j_{k}})$ is not statistically $p$-quasi-Cauchy. Then the sequence $(\alpha_{j})$ does not have any statistically $p$-quasi-Cauchy subsequence. If $A$ is bounded above and unbounded below, then pick an element $\beta_{1}$ of $A$ less than $-p$. Then we can find an element $\beta_{2}$ of $A$ such that $\beta_{2}<-2p+\beta_{1}$. Similarly, choose an element $\beta_{3}$ of $A$ such that $\beta_{3}<-3p+\beta_{2}$. Thus one can construct a sequence $(\beta_{i})$ of points in $A$ such that  $\beta_{i+1}<-(i+1)p+\beta _{i}$ for each $i\in{\mathbb{N}}$. Let $(\beta _{i_{k}})$ be any subsequence of $(\beta _{i})$. Since $\{k\leq n: |\beta _{i_{k+1}}-\beta _{i_{k}}|\geq p\}=\{1, 2, ..., n\}$, it follows that
$\lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq n: |\beta _{i_{k+1}}-\beta _{i_{k}}|\geq p\}|=1\neq{0}$. This implies that $(\beta _{i_{k}})$ is not statistically $p$-quasi-Cauchy. Then the sequence $(\alpha_{i})$ does not have any statistically $p$-quasi-Cauchy subsequence.
Thus this contradiction completes the proof of the theorem.
\end{pf}
It follows from Theorem \ref{ThmEisboundediffitispwardcompact} that statistically $p$-ward compactness of a subset of $A$ of $\mathbb{R}$ coincides with either of the following kinds of compactness: $p$-ward compactness (\cite[Theorem 2.3]{CakalliVariationsonquasiCauchysequences}), statistical ward compactness (\cite[Lemma 2]{CakalliStatisticalwardcontinuity}), $\lambda$-statistical ward compactness (\cite[Theorem 1]{CakalliandSonmezandAraslamdastatisticallywardcontinuity}), $\rho$-statistical ward compactness (\cite[Theorem 1]{CakalliAvariationonstatisticalwardcontinuity}), strongly lacunary ward compactness (\cite[Theorem 3.3]{CakalliNthetawardcontinuity}), slowly oscillating compactness (\cite[Theorem 3]{CakalliStatisticalquasiCauchysequences}), lacunary statistical ward compactness (see \cite{CakalliandArasandSonmezLacunarystatisticalwardcontinuity}), and \cite[Theorem 3]{CakalliStatisticalquasiCauchysequences}), ideal ward compactness (\cite[Theorem 8]{CakalliandHazarikaIdealquasiCauchysequences}), Abel ward compactness (\cite[Theorem 5]{CakalliandAlbayrakNewtypecontinuitiesviaAbelconvergence}).

If a closed subset of $\mathbb{R}$ is statistically $p$-ward compact for a positive integer $p$, then any sequence of points in $A$ has a $(P_n ,s)$-absolutely almost convergent subsequence (see \cite{CakalliGCanakPnsabsolutealmostconvergentsequences}, \cite{CakalliandTaylanOnabsolutelyalmostconvergence}, \cite{OzarslanandYildizAnewstudyontheabsolutesummabilityfactorsofFourierseries}, \cite{YildizAnewtheoremonlocalpropertiesoffactoredFourierseries}, \cite{BorOnGeneralizedAbsoluteCesaroSummability}, and \cite{YildizOnAbsoluteMatrixSummabilityFactorsofInfiniteSeriesandFourierSeries}).

\begin{cor}
\normalfont
A subset of $\mathbb{R}$ is statistically $p$ ward compact if and only if it is statistically $q$ ward compact for any $p, q\in {\mathbb{N}}$.
\end{cor}

\begin{cor}
\normalfont
A subset of $\mathbb{R}$ is statistically $p$ ward compact if and only if it is both statistically upward half compact and statistically downward half compact.
\end{cor}
\begin{pf}
The proof follows from \cite[Corollary 3.9]{CakalliUpwardanddownwardstatisticalcontinuities}, so is omitted.
\end{pf}

\begin{cor}
\normalfont
A subset of $\mathbb{R}$ is  statistically  $p$ ward compact for a $p\in{\mathbb{N}}$ if and only if it is both lacunary statistically upward half compact and lacunary statistically downward half compact.
\end{cor}
\begin{pf}
The proof follows from \cite[Theorem 1.3 and Theorem 1.9]{CakalliandMucukLacunarystatisticallyupwardanddownwardhalfquasiCauchysequencesJofAnalysis}, so is omitted.
\end{pf}

\section{Variations on statistical ward continuity}
 In this section, we investigate connections between uniformly continuous functions and statistically $p$-ward continuous functions. A function $f:\mathbb{R} \longrightarrow \mathbb{R}$ is continuous if and only if it preserves statistically convergent sequences. Using this idea, we introduce statistical  $p$-ward continuity.
\begin{Defn}
A function $f$ is called statistically $p$-ward continuous on a subset $A$ of $\mathbb{R}$ if it preserves statistically $p$-quasi-Cauchy sequences, i.e. the sequence $f(\textbf{x})=(f(\alpha_{n}))$ is statistically $p$-quasi-Cauchy whenever $\boldsymbol\alpha=(\alpha_{n})$ is statistically $p$-quasi-Cauchy of points in $A$.
\end{Defn}
We see that this definition of statistically $p$-ward continuity can not be obtained by any summability matrix $A$ (see \cite{ConnorandGrosseSequentialdefinitionsofcontinuityforrealfunctions}).

We note that the sum of two statistically $p$-ward continuous functions is statistically $p$-ward continuous, and for any constant $c\in \mathbb{R}$, $cf$ is statistically $p$-ward continuous whenever $f$ is a statistically $p$-ward continuous function, so that the set of all statistically $p$ ward continuous functions is a vector space. The composite of two statistically $p$-ward continuous functions is statistically $p$-ward continuous, but the product of two statistically $p$-ward continuous functions need not be statistically $p$-ward continuous as it can be seen by considering product of the statistically $p$-ward continuous function $f(x)=x$ with itself. If $f$ is a statistically $p$-ward continuous function, then $|f|$ is also statistically $p$-ward continuous since
$$|\{k\leq n:  |f(\alpha_{k+p})-f(\alpha_{k})|\geq{\varepsilon}\}|\subseteq |\{k\leq n: ||f(\alpha_{k+p})|-|f(\alpha_{k})||\geq{\varepsilon}\}|$$
which follows from the inequality $||f(\alpha_{k+p})|-|f(\alpha_{k})|| \leq  |f(\alpha_{k+p})-f(\alpha_{k})|$.
%for every $n\in {\mathbb{N}}$.
If $f$, and $g$ are statistically $p$-ward continuous, then $max\{f, g\}$ is also statistically $p$-ward continuous, which follows from the equality $max\{f, g\}=\frac{1}{2}\{|f-g|+|f+g|\}$.

In connection with statistically $p$-quasi-Cauchy sequences, slowly oscillating sequences, and convergent sequences the problem arises to investigate the following types of  �continuity� of a function on $\mathbb{R}$.
\begin{description}
\item[($\Delta^{s}_{p} $)] $(\alpha_{n}) \in {\Delta^{s}_{p}} \Rightarrow (f(\alpha_{n})) \in {\Delta^{s}_{p}}$
\item[($\Delta^{s}_{p} c$)] $(\alpha_{n}) \in {\Delta^{s}_{p}} \Rightarrow (f(\alpha_{n})) \in {c}$
\item[$(\Delta^{s})$] $(\alpha_{n}) \in {\Delta^{s}} \Rightarrow (f(\alpha_{n})) \in {\Delta^{s}}$
\item[$(c)$] $(\alpha_{n}) \in {c} \Rightarrow (f(\alpha_{n})) \in {c}$
\item[$(d)$] $(\alpha_{n}) \in {c} \Rightarrow (f(\alpha_{n})) \in {\Delta^{s}_{p}}$
\item[$(e)$] $(\alpha_{n}) \in {w} \Rightarrow (f(\alpha_{n})) \in {\Delta^{s}_{p}}$
\end{description}
where $w$ denotes the set of slowly oscillating sequences, and $\Delta^{s}=\Delta^{s}_{1}$. We see that $(\Delta^{s}_{p})$ is statistically $p$-ward continuity of $f$, and $(c)$ states the ordinary continuity of $f$. It is easy to see that $(\Delta^{s}_{p} c)$ implies $(\Delta^{s}_{p})$, and $(\Delta^{s}_{p} )$ does not imply $(\Delta^{s}_{p} c)$;  $\Delta^{s}_{p}$ implies $(d)$, and $(d)$ does not imply $(\Delta^{s}_{p})$;  and $(\Delta^{s}_{p})$ implies $(e)$, and $(e)$ does not imply $(\Delta^{s}_{p})$; $(\Delta^{s}_{p}c)$ implies $(c)$ and $(c)$ does not imply $(\Delta^{s}_{p} c)$; and $(c)$ implies $(d)$.
Now we see that $(\Delta^{s}_{p})$ implies $(\Delta^{s})$, i.e. any statistically $p$-ward continuous function is statistically ward continuous.
\begin{thm} \label{TheopquasiCauchyimpliesquasiCauchy}   If $f$ is statistically $p$-ward continuous on a subset $A$ of $\mathbb{R}$ for some $p\in{\mathbb{N}}$, then it is statistically ward continuous on $A$.
\end{thm}
\begin{pf}  If $p=1$, then it is obvious. So we would suppose that $p>1$. Take any statistically $p$-ward continuous function $f$ on $A$. Let $(\alpha_{k})$ be any statistical quasi-Cauchy sequence of points in $A$. Write $$(\xi_{i})=(\alpha_{1}, \alpha_{1},..., \alpha_{1}, \alpha_{2}, \alpha_{2},..., \alpha_{2},..., \alpha_{n}, \alpha_{n},..., \alpha_{n}, ...),$$ where the same term repeats $p$ times.
The sequence $$(\alpha_{1}, \alpha_{1},..., \alpha_{1}, \alpha_{2}, \alpha_{2},..., \alpha_{2},..., \alpha_{n}, \alpha_{n},..., \alpha_{n}, ...)$$ is also statistically quasi-Cauchy so it is statistically $p$-quasi-Cauchy. By the statistically $p$-ward continuity of $f$, the sequence
$$(f(\alpha_{1}), f(\alpha_{1}), ... ,f(\alpha_{1}),f(\alpha_{2}), f(\alpha_{2}),..., f(\alpha_{2}),..., f(\alpha_{n}), f(\alpha_{n}), ..., f(\alpha_{n}), ...)$$ is statistically $p$-quasi-Cauchy, where the same term repeats $p$-times.
Thus the sequence $$(f(\alpha_{1}), f(\alpha_{1}),..., f(\alpha_{1}), f(\alpha_{2}), f(\alpha_{2}),..., f(\alpha_{2}),..., f(\alpha_{n}), f(\alpha_{n}),..., f(\alpha_{n}), ...)$$ is also statistically $p$ quasi-Cauchy. It is easy to see that
$s_{t}-lim (f(\alpha_{n+p})-f(\alpha_{n}))=0$, which completes the proof of the theorem.

\end{pf}
\begin{cor}   If $f$ is statistically $p$-ward continuous on a subset $A$ of $\mathbb{R}$, then it is continuous on $A$ in the ordinary case.
\end{cor}
\begin{pf} The proof follows immediately from \cite[Theorem 3]{CakalliStatisticalwardcontinuity} so is omitted.
\end{pf}
\begin{thm} Statistical $p$-ward continuous image of any statistically $p$-ward compact subset of $\mathbb{R}$ is statistically $p$-ward compact.
\end{thm}
\begin{pf}
Let $f$ be a statistically $p$-ward continuous function, and $A$ be a statistically $p$-ward compact subset of $\mathbb{R}$. Take any sequence $\boldsymbol \beta=(\beta_{n})$ of terms in $f(E)$. Write $\beta_{n}=f(\alpha_{n})$ where $\alpha_{n}\in {E}$ for each $n \in{\mathbb{N}}$, $\boldsymbol\alpha=(\alpha_{n})$. statistically $p$-ward compactness of $A$ implies that there is a statistically $p$-quasi-Cauchy subsequence
$\boldsymbol \xi=(\xi_{k})=(\alpha_{n_{k}})$ of $\boldsymbol\alpha$. Since $f$ is statistically $p$-ward continuous, $(t_{k})=f(\boldsymbol \xi)=(f(\xi_{k}))$ is statistically $p$-quasi-Cauchy. Thus $(t_{k})$ is a statistically $p$-quasi-Cauchy subsequence of the sequence $f(\boldsymbol\alpha)$. This completes the proof of the theorem.
\end{pf}
\begin{cor} Statistical $p$-ward continuous image of any $G$-sequentially connected subset of $\mathbb{R}$ is $G$-sequentially connected for a regular subsequential method $G$ (see \cite{CakalliOnGcontinuity}, \cite{MucukandCakalliGsequentiallyconnectednessfortopologicalgroupswithoperations}, and \cite{CakalliandMucukconnectednessviaasequentialmethod}).
\end{cor}
\begin{pf} The proof follows from the preceding theorem, so is omitted (see \cite{CakalliSequentialdefinitionsofconnectedness} and \cite{MucukandSahanOnGsequentialcontinuity} for the definition of $G$-sequential connectedness and related concepts).
\end{pf}
\begin{thm}  \label{TheouniformcontinuityimpliesthatfalpahansistatisticallypquasiCauchywheneveralphanispquasiCauchy} If $f$ is uniformly continuous on a subset $A$ of $\mathbb{R}$, then $(f(\alpha_{n}))$ is statistically $p$-quasi-Cauchy whenever $(\alpha_{n})$ is a $p$-quasi-Cauchy sequence of points in $A$.
\end{thm}
\begin{pf}
Let $(\alpha_{n})$ be any $p$-quasi-Cauchy sequence of points in $A$. Take any $\varepsilon > 0$. Uniform continuity of $f$ on $A$ implies that there exists a $\delta >0$, depending on $\varepsilon$, such that $|f(x)-f(y)|< \varepsilon$ whenever  $|x-y|< \delta$ and $x, y\in{A}$. For this $\delta >0$, there exists an $N=N(\delta)$ such that $|\Delta_{p} \alpha_{n}|<\delta$ whenever $n>N$. Hence $|\Delta_{p} f(\alpha_{n})|<\varepsilon$ if $n>N$. Thus $\{k\leq n: |\Delta_{p} f(\alpha_{k})|\geq \varepsilon\}\subseteq \{1, 2, ..., N\}$. Therefore
$\lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq n:|\Delta_{p} f(\alpha_{k})|\geq \varepsilon\}|\leq \lim_{n\rightarrow\infty}\frac{1}{n}|\{k\leq N:k\in{\mathbb{N}}\}|=0$.
It follows from this that $(f(\alpha_{n}))$ is a statistically $p$-quasi-Cauchy sequence. This completes the proof of the theorem.
\end{pf}
\begin{cor} If $f$ is slowly oscillating continuous on a bounded subset $A$ of $\mathbb{R}$, then $(f(\alpha_{n}))$ is statistically $p$-quasi-Cauchy whenever $(\alpha_{n})$ is a $p$ quasi-Cauchy sequence of points in $A$.
\end{cor}
\begin{pf}
If $f$ is a slowly oscillating continuous function on a bounded subset $A$ of $\mathbb{R}$, then it is uniformly continuous on $A$ by \cite[Theorem 2.3]{CanakandDikNewtypesofcontinuities}. Hence the proof follows from Theorem \ref{TheouniformcontinuityimpliesthatfalpahansistatisticallypquasiCauchywheneveralphanispquasiCauchy}.
\end{pf}
It is well-known that any continuous function on a compact subset $A$ of $\mathbb{R}$ is uniformly continuous on $A$. We have an analogous theorem for a statistically $p$-ward continuous function defined on a statistically $p$-ward compact subset of $\mathbb{R}$.
\begin{thm}
If a function is statistically $p$-ward continuous on a statistically $p$-ward compact subset of $\mathbb{R}$, then it is uniformly continuous on $A$.
\end{thm}
\begin{pf}
Suppose that $f$ is not uniformly continuous on $A$ so that there exist an $\epsilon _{0}>0$ and sequences $(\alpha_{n})$ and $(\beta_{n})$ of points in $A$ such that $|\alpha_{n}-\beta_{n}|<1/n$ and $|f(\alpha_{n})-f(\beta_{n})|\geq \epsilon _{0}$ for all $n \in \mathbb{N}$. Since $A$ is statistically $p$-ward compact, there is a subsequence $(\alpha_{n_{k}})$ of $(\alpha_{n})$ that is statistically $p$-quasi-Cauchy. On the other hand, there is a subsequence $(\beta_{n_{k_{j}}})$ of $(\beta_{n_{k}})$ that is statistically $p$-quasi-Cauchy as well. It is clear that the corresponding sequence $(a_{n_{k_{j}}})$ is also statistically $p$-quasi-Cauchy, since \\
$\{j\leq n: |\alpha_{n_{k_{j+p}}}-\alpha_{n_{k_{j}}}|\geq{\varepsilon}\}\subseteq \{j\leq n: |\alpha_{n_{k_{j+p}}}-\beta_{n_{k_{j+p}}}|\geq{\frac{\varepsilon}{3}}\} \cup \{j\leq n: |\beta_{n_{k_{j+p}}}-\beta_{n_{k_{j}}}|\geq{\frac{\varepsilon}{3}}\}\cup \{j\leq n: |\beta_{n_{k_{j}}}-\alpha_{n_{k_{j}}}|\geq{\frac{\varepsilon}{3}}\}$\\
for every $n\in {\mathbb{N}}$, and for every $\varepsilon>0$.  Now the sequence \\$$(\alpha_{n_{k_{1}}},\alpha _{n_{k_{1}}},..., \alpha_{n_{k_{1}}}, \beta_{n_{k_{1}}}, \beta_{n_{k_{1}}},... ,\beta_{n_{k_{1}}},  ..., \alpha_{n_{k_{j}}}, \alpha_{n_{k_{j}}},...,\alpha_{n_{k_{j}}}, \beta_{n_{k_{j}}}, \beta_{n_{k_{j}}}, ..., \beta_{n_{k_{j}}},...)$$ is statistically $p$-quasi-Cauchy while the sequence
$$(f(\alpha_{n_{k_{1}}}),f(\alpha _{n_{k_{1}}}),...,f(\alpha_{n_{k_{1}}}),f(\beta_{n_{k_{1}}}),f(\beta_{n_{k_{1}}}),... ,f(\beta_{n_{k_{1}}}),  ...,f(\alpha_{n_{k_{j}}}),f(\alpha_{n_{k_{j}}}),...,f(\alpha_{n_{k_{j}}}), f(\beta_{n_{k_{j}}}),f(\beta_{n_{k_{j}}}),...,f(\beta_{n_{k_{j}}}),...)$$
is not statistically $p$-quasi-Cauchy where same term repeats $p$-times. Hence this establishes a contradiction, so completes the proof of the theorem.
\end{pf}
\begin{cor} If a function defined on a bounded subset of $\mathbb{R}$ is statistically $p$-ward continuous, then it is uniformly continuous.
\end{cor}
We note that when the domain of a function is restricted to a bounded subset of $\mathbb{R}$, statistically $p$-ward continuity implies not only ward continuity, but also slowly oscillating continuity.
\section{Conclusion}
In this paper, we introduce statistically $p$-quasi Cauchy sequences, and investigate conditions for a statistically $p$ ward continuous real function to be uniformly continuous, and prove some other results related to these kinds of continuities and some other kinds of continuities. It turns out that statistically $p$-ward continuity implies uniform continuity on a bounded subset of $\mathbb{R}$.  The results in this paper not only involves the related results in \cite{CakalliStatisticalwardcontinuity} as a special case for $p=1$, but also some interesting results which are also new for the special case $p=1$. The statistically $p$-quasi Cauchy concept for $p>1$ might find more interesting applications than statistical quasi Cauchy sequences to the cases when statistically quasi Cauchy does not apply. For a further study, we suggest to investigate statistically $p$-quasi-Cauchy sequences of soft points and statistically $p$-quasi-Cauchy sequences of fuzzy points. However due to the change in settings, the definitions and methods of proofs will not always be analogous to those of the present work (for example see \cite{ArasandSonmezandCakalliAnapproachtosoftfunctions}, \cite{CakalliandPratulFuzzycompactnessviasummability}, \cite{ErdemandArasandCakalliandSonmezSoftmatricesonsoftmultisetsinanoptimaldecisionprocess}, and  \cite{KocinacSelectionpropertiesinfuzzymetricspaces}). We also suggest to investigate statistically $p$-quasi-Cauchy double sequences of points in $\mathbb{R}$ (see \cite{PattersonandSavasAsymptoticequivalenceofdoublesequences}, \cite{{PattersonandCakalliQuasiCauchydoublesequences}}, \cite{DjurcicandKocinacandZizovicDoublesequencesandselections}, and \cite{CakalliandPattersonFunctionspreservingslowlyoscillatingdoublesequences} for the related definitions in the double case). For another further study, we suggest to investigate statistically $p$-quasi-Cauchy sequences in abstract metric spaces (see \cite{CakalliandSonmezandGencOnanequivalenceoftopologicalvectorspacevaluedconemetricspacesandmetricspaces}, \cite{PalandSavasandCakalliIconvergenceonconemetricspaces}, \cite{CakalliandSonmezSlowlyoscillatingcontinuityinabstractmetricspaces}, \cite{SonmezOnparacompactnessinconemetricspaces}, and  \cite{YayingandHazarikandCakalliNewresultsinquasiconemetricspaces}).





\end{document}
